Author's response to reviews

Title: Two common nonsynonymous paraoxonase 1 (PON1) gene polymorphisms and brain astrocytoma and meningioma

Authors:

Carmen Martinez (cmartine@unex.es)
Jose A Molina (jamolina.h12o@salud.madrid.org)
Hortensia Alonso-Navarro (hortalon@yahoo.es)
Félix Javier Jiménez-Jiménez (fjavier.jimenez@salud.madrid.org)
José A Agúndez (jagundez@unex.es)
Elena Garcia-Martin (elenag@unex.es)

Version: 5 Date: 19 August 2010

Author's response to reviews:

All formatting changes were made as requested.